Author’s response to reviews

Title: Process evaluation of the response of nursing homes to the implementation of the dementiaspecific case conference concept WELCOME-IdA: A qualitative study

Authors:

Daniela Holle (daniela.holle@hs-gesundheit.de)
Sonja Teupen (sonja.teupen@dzne.de)
Rabea Graf (rabea.graf@uni-bielefeld.de)
Rene Müller-Widmer (rene.mueller-widmer@dzne.de)
Sven Reuthen (sven.reuther@dzne.de)
Margareta Halek (margareta.halek@dzne.de)
Martina Roes (martina.roes@dzne.de)

Version: 4 Date: 30 Jan 2020

Author’s response to reviews:

Dear Editor and Reviewer,

Thank you very much for your valuable feedback and comments regarding our manuscript. Below you will find the point-by-point responses to your feedback and recommendations.

Kind regards,

Daniela Holle, on behalf of all authors

Editor’s comments

1. Please ensure that all additional files are explicitly referred to in the main text. Any items which do not meet these requirements may be deleted by our production department.

Revision

We checked that all additional files are referred to in the main text.
Editor’s comments

2. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Revision

We removed any tracked changes or highlighting and included a single clean copy of the manuscript.